### RESEARCH

**Reproductive Health** 



# A bit of medical paternalism? A qualitative study on power relations between women and healthcare providers when deciding on mode of birth in five public maternity wards of Argentina

M. Vila Ortiz<sup>1,3\*</sup>, C. Gialdini<sup>2,3</sup>, C. Hanson<sup>1</sup>, A. P. Betrán<sup>4</sup>, G. Carroli<sup>3</sup> and H. Mølsted Alvesson<sup>1</sup>

### Abstract

**Background** Whether women should be able to decide on mode of birth in healthcare settings has been a topic of debate in the last few decades. In the context of a marked increase in global caesarean section rates, a central dilemma is whether pregnant women should be able to request this procedure without medical indication. Since 2015, Law 25,929 of Humanised Birth is in place in Argentina. This study aims at understanding the power relations between healthcare providers, pregnant women, and labour companions regarding decision-making on mode of birth in this new legal context. To do so, central concepts of power theory are used.

**Methods** This study uses a qualitative design. Twenty-six semi-structured interviews with healthcare providers were conducted in five maternity wards in different regions of Argentina. Participants were purposively selected using heterogeneity sampling and included obstetrician/gynaecologists (heads of department, specialists working in 24-h shifts, and residents) and midwives where available. Reflexive thematic analysis was used to inductively develop themes and categories.

**Results** Three themes were developed: (1) Healthcare providers reconceptualize decision-making processes of mode of birth to make women's voices matter; (2) Healthcare providers feel powerless against women's request to choose mode of birth; (3) Healthcare providers struggle to redirect women's decision regarding mode of birth. An overarching theme was built to explain the power relations between healthcare providers, women and labour companions: Healthcare providers' loss of *beneficial power* in decision-making on mode of birth.

**Conclusions** Our analysis highlights the complexity of the healthcare provider-woman interaction in a context in which women are, in practice, allowed to choose mode of birth. Even though healthcare providers claim to welcome women being an active part of the decision-making processes, they feel powerless when women make autonomous decisions regarding mode of birth. They perceive themselves to be losing *beneficial power* in the eyes of patients and consider fruitful communication on risks and benefits of each mode of birth to not always be possible. At the same time, providers perform an increasing number of CSs without medical indication when it is convenient for them, which suggests that paternalistic practices are still in place.

\*Correspondence: M. Vila Ortiz mercedes.vila.ortiz@ki.se Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/A.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

Content courtesy of Springer Nature, terms of use apply. Rights reserved.

Keywords Mode of birth, Decision-making, Caesarean section, Power relations, Medical paternalism

#### Plain Language Summary

In the last few decades, there has been a debate on whether women should be able to choose if they haver a vaginal birth or a caesarean section. This debate has been framed by the fact that an increasing number of caesarean sections are being performed. Since 2015, Argentina has a Law of Humanised Birth. We conducted a study to understand the power relations between healthcare providers, pregnant women and labour companions in decision making on mode of birth in this new legal context. To do so, we used central concepts of power theory. We conducted 26 semi-structured interviews with healthcare providers in five maternity wards of Argentina. The interviewees were obstetrician/gynaecologists (heads of department, specialists working in 24-h shifts, and residents) and midwives where available. We used thematic analysis to build themes from the data. We discovered that healthcare providers providers perceive themselves to be losing *beneficial power* in decision-making on mode of birth. Even though they claim to want women to make autonomous decisions, they feel frustrated when this happens. They also perceive it to be more difficult to communicate with patients regarding the risks and benefits of vaginal birth and caesarean section. At the same time, providers carry out an increasing number of CSs without medical indication when it is convenient for them, which suggests that paternalistic practices are still in place.

#### Background

Women being able to decide on mode of birth in healthcare settings has been a topic of debate in the last few decades, giving rise to arguments both in favour and against this practice [1–9]. Self-determination movements have emphasised women's right to make decisions regarding their health, supporting their reasoning with the principle of patient autonomy [2, 4, 5, 10, 11]. Those who argue against giving women the right to choose mode of birth highlight the importance of evidence-based practice and the autonomy of physicians, suggesting that patients should not be able to demand a treatment that healthcare providers (HCPs) do not consider appropriate [1, 6] and for which they are legally responsible.

In the context of a marked increase in global caesarean section (CS) rates [12, 13], a central dilemma is whether or not pregnant women should be able to choose a CS without medical indication as mode of birth. CS is a life-saving intervention when clinically indicated. However, the available evidence suggests that carrying out this procedure without clinical indication is associated with increased risks for short and long-term adverse outcomes [14]. Moreover, it has been pointed out that CS upon request diverts resources [1, 15] and places further economic burden in health systems [16].

Decision-making in healthcare settings involves building relations between women and healthcare providers. The relational nature of care has been a long-standing field of study [17]. Recent evidence suggests that HCPs in maternity wards generally welcome women's increased involvement in the birth process and in decision-making on mode of birth, including CS upon maternal request, especially in high-income countries [7]. Previous research on the topic has focused on the importance of the communication and trust aspects of decisionmaking, women's and providers' values and preferences regarding mode of birth and provision of information and support for women, among others [7, 18–20]. Furthermore, feminist theories have produced an important body of knowledge on the politics of birth [21–23]. However, few empirical studies have focused their attention on how power relations between HCPs and women, and the legal framework in which they develop, shape the way decisions regarding mode of birth are made in healthcare settings [24–26].

# Theoretical framework: How do we understand power relations in maternity care?

In the biomedical model of care, described in Table 1, the traditional patient-doctor relationship is asymmetrical. HCPs are legally and institutionally authorised to make medical decisions [27] and usually decide which treatment options patients are offered [28, 29]. Moreover, as they are considered the legitimate knowledge authority [30], HCPs generally control the terms in which health issues are discussed, which is an important dimension of power in decision-making [31]. This is also the case for the HCP-pregnant woman relationship in maternal health [29]. A study conducted in Australia has shown, for example, how paternalistic practices contribute to women's lack of control during childbirth [25].

However, power relations established within the biomedical model of pregnancy are not static. Any power dynamics between HCPs and patients involve negotiation and resistance as well as dominance in the medical encounter [17]. In the neoliberal model of medical care, patients in high-income countries have shifted from being considered mere 'recipients' of treatment

#### Table 1 The biomedical model of care in maternal health

#### The biomedical model of care in maternal health

• This model was established as a result of birth shifting from a domestic practice led by women to an institutionalized event handled by HCPs in increasingly functional health systems during the 19th and 20th century [22, 34]

• It is characterised by:

• The medicalization of women's body [35] and their reproductive functions [32, 36, 37], which translates to pregnancy and birth being treated like a disease rather than a physiological process. The pathologization of birth was criticised by feminist theories and the feminist movement as it was considered to legitimise the expansion of medical control at the expense of women's agency and preferences [22]

• The separation of the biological aspects of reproductive processes from the emotional and subjective ones [38]

• Pregnancy and birth controlled by HCPs in a clinical environment, with a doctor-centred care and decision-making process in maternity wards [39]. This aspect of the biomedical model has been challenged in some high-income countries, in which birth is assisted by different cadres—for example, midwives—and with several concurrent degrees of institutionalization and medicalization [40]

to 'consumers' or even 'reflexive consumers' of healthcare, suggesting that they more actively participate in the care they receive [32]. In the last few decades, initiatives towards a patient-centred model of care that gives women more control over maternal health have also been proposed [10, 28, 33].

In 2020, Eide and Bærøe, researchers at the University of Bergen, Norway, suggested a concept to structure how decisions regarding mode of birth are reached in maternity wards: *Beneficial power* [29]. It refers to HCPs being able to influence pregnant women's choice of the best clinical option for them, vaginal birth (VB) or CS, by taking into consideration patients' own perceptions of needs and desires, especially in the case of CS upon maternal request.

However, this definition of power only works when the doctor-patient relationship is built upon patients' trusting and valuing HCPs' opinions and, in turn, HCPs involving patients in the process of decision-making. What happens when this is not the case? Eide and Bærøe address the *opposing autonomous claims* that arise when women ask for a caesarean section without medical indication, which can lead to an *experience of powerlessness* for both providers and patients [29].

In 2004, Law n. 25,929 of Humanized Birth was approved in Argentina. It was regulated in 2015, following the pressure of respectful birth movements to improve the quality of care. This law states the rights of women, families and babies during pregnancy, labour, birth, and postpartum period in healthcare facilities. The law includes women's right to be treated with respect, to participate in the decision-making process during labour and birth, to be informed about the medical interventions she may receive, to have a companion of choice, to make an informed and free decision regarding where and how to go to labour, among others [41]. There is limited evidence as to how this new law has affected the way in which mode of birth is decided upon in health facilities [42, 43]. In this paper we address the decision-making

processes and the interaction between doctors, pregnant women and labour companions when deciding on the mode of birth in a new legal context. Our study reports a qualitative analysis of a formative research about CS determinants carried out in 2018–2019 in Argentina [44]. This is a middle-income country that has seen an increase in rates of CS (34.7% in 2017, according to the National Perinatal Information System), including CS on maternal request [45], during the last few decades. The aim of this qualitative study is to understand the power relations between HCPs, pregnant women, and labour companions in decision-making on mode of birth in hospitals of Argentina from the perspective of HCPs.

#### Methods

#### Setting

This study was conducted in public maternity wards in Argentina. The health system in this country comprises the public health system, the social security system, and the private health system. The public system is available for anyone who is in Argentinean territory free of charge. The social security system is made up of health insurance coverage managed by trade unions, provincial governments, and other institutions (such as Universities). It is mainly financed by personal contributions from workers and employers, voluntary contributions, co-payments, benefit contributions from the health system, etc. Finally, private companies constitute the private sector.

The majority of births occur in healthcare facilities (99.7%), predominantly in public hospitals (62.5%) [46]. In the public health system, healthcare facilities and providers receive the same payment irrespective of the type of birth, and patients are not charged when receiving care. The national legislation regulates protocols to manage labour and birth. However, there is no specific policy on the use of CS.

#### Design

In this qualitative study, we used 26 semi-structured interviews with HCPs. The interviews were carried out during October 2018-August 2019 in a sub-sample of five hospitals as part of a larger mixed-method formative research project in nineteen public hospitals in Argentina, aimed at exploring key informants' view of the feasibility of implementing non-clinical interventions to reduce caesarean section rates [44]. This formative research was commissioned by the World Health Organization conducted by Centro de Estudios de Estado y Sociedad (CEDES) and Centro Rosarino de Estudios Perinatales (CREP). Results of the mixed-method study have been published elsewhere [47, 48] and have been used to inform the adaptation of non-clinical interventions to optimize the use of CS within the study "Quality decision-making (QUALI-DEC) by women and healthcare providers for appropriate use of caesarean section" [46]. Here we present further in-depth analysis of the qualitative interviews conducted during the formative research.

#### Data collection and participants

Five public maternity services in five different jurisdictions of Argentina were selected from the nineteen hospitals participating in the formative research as a subsample to carry out in-depth interviews. The criteria to participate in the formative research included using an electronic system to register maternal and perinatal medical records and having more than 1000 births per year. The nineteen facilities were purposively selected from all the eligible hospitals (88 in total) to cover different geographical regions, and for variation in social and economic development as well as health outcomes across the five regions of the country [49]. The sub-sample of five hospitals in which the in-depth interviews were carried out are in the provinces of Santa Fe, Salta, Corrientes, Tucumán and in Buenos Aires City. All participating hospitals provide access to emergency CS and have resources to handle obstetric emergencies [48].

Semi-structured interviews were carried out [50]. The interview guide consisted in 16 open questions and explored the determinants of choice regarding mode of birth, especially focusing on caesarean section births, and key actors' perceptions regarding the usefulness and feasibility of non-clinical interventions to optimize the use of CS. Two experienced researchers carried out the interviews either in person or virtually, according to the convenience and preference of the interviewees. The interviews had a duration of 30 to 60 min and were recorded.

In each hospital, the participants were purposively selected using heterogeneity sampling regarding professions and positions in the healthcare facilities [44]. Twenty-six HCPs working in five provinces of Argentina were interviewed. Details about jurisdiction, region, and type of provider are presented in Table 2. In hospitals that had midwives in their staff, these were included in the sample. Among obstetricians, the sample included heads of department, specialists working in 24-h shifts, and residents. To recruit participants, researchers from CEDES and CREP contacted a focal person in each hospital, who in turn asked HCPs if they would be interested in participating in the study. Those that accepted to be interviewed were then contacted by a member of the research team.

#### Data analysis

We conducted an analysis to explore the power dynamics between women, companions, and HCPs in the mode of birth decision-making process from the perspective of HCPs. The interviews were coded using Nvivo. Reflexive thematic analysis [51, 52] was used to inductively build themes and categories, addressing both semantic and latent content. The first author, MVO, led the coding process with multiple discussions with HMA and CG, who is a member of the research team with strong contextual knowledge who was part of the team conducting the formative research. Consensus throughout the codes was not assumed, so the first author led a process

#### Table 2 Participating maternity hospitals and types of HCPs

Hospital	Jurisdiction	Region	Types of HCPs	Total of interviews per hospital
Hospital 1	Tucumán	Northwest	3 Obstetricians, 2 midwives	5
Hospital 2	Salta	Northwest	2 Obstetricians, 1 obstetric resident, 2 midwives	5
Hospital 3	Santa Fe	Centre	4 Obstetricians, 1 obstetric resident	5
Hospital 4	Corrientes	Litoral	4 Obstetricians, 1 obstetric resident	5
Hospital 5	Ciudad Autónoma de Buenos Aires	Centre	4 Obstetricians, 1 obstetric resident, 1 midwife	6

of comparing and contrasting codes together with an insider (CG) and an outsider researcher (HMA). Following several rounds of discussions, we decided how to best develop the themes. Themes were discussed with all co-authors. During the analysis, we compared the five participating hospitals and built crosscutting themes that highlight the commonalities among them regarding decision-making on mode of birth. Therefore, this should be considered a single-case study. The analysis was conducted by an interdisciplinary team made up by researchers with contextual knowledge and expertise in global health.

#### Results

Three main themes were developed from the analysis, which are synthesised in Table 3, with their corresponding categories.

### Theme 1: HCPs reconceptualize decision-making processes of mode of birth

Changes in how HCPs conceptualize the decision-making processes regarding mode of birth was emphasized. HCPs claim to be changing their attitudes when deciding between a VB or a CS in the healthcare setting, shifting from a doctor-centred towards a patient-centred model of care. In their view, the former 'hegemonic' biomedical model of care was characterised by medical interventionism and doctors' decision-making. The new model, which is not yet completely instituted, is based in women's rights and acknowledges patients' agency to make decisions during the labour and birth process.

The change is usually described using temporal marks like *before* versus *now*, distinguishing a somehow relatively distant past versus current values. Interviewees state that nowadays women's opinions and wishes are taken into consideration when determining mode of birth in healthcare facilities and that woman are encouraged to take an active role in the process:

For me, today (...) the power to choose how to give birth lies with the woman, as we have made the slogan of the week of respectful childbirth, to choose how to give birth (Hospital 1, midwife).

The shift in decision-making conceptualization appears closely related to CS by women's request. Even though interviewed HCPs agree that CS by maternal request cannot account for the high CS rates in their facilities, it has become an important factor in the doctor-patient interaction regarding mode of birth decision-making.

One thing that we take into account now and that we did not before is the will of the patient. Before, if a patient told you 'I want a caesarean section', we almost did not listen to her, and now it is not that way (Hospital 3, obstetrician).

The roots of the change were identified both *within* and *without* the health system. Internal changes include HCPs realising that mode of birth is not something that can be imposed by doctors, and that women's wishes and expectations should be listened to and taken into consideration. As a result, providers are trying to have a less interventionist and paternalist mindset when it comes to labour and birth: 'We have changed'.

What I have noticed over the years, what has changed the most is that sometimes the patient has more information, and today the patient is listened to much more. We have all changed as professionals; the patient's wish for a caesarean section is respected much more than before. Before, she would say, 'I want a caesarean section', and one applied a

Table 3	Themes and	categories

Overarching theme	HCPs loss of beneficial power in decision-making on mode of birth			
Theme one	HCPs reconceptualize decision-making processes of mode	Temporality		
	of birth to make women's voices matter	Before: Doctor-centred decision-making		
		Now: towards a patient-centred decision-making		
		Spatiality		
		Changes from inside: change in HCPs attitudes		
		Changes from outside: changes in legislation and women's attitudes		
Theme two	HCPs feel powerless against women's request to choose mode of birth	HCPs frustration when women ask for a CS or birth plan HCPs intent on controlling the process HCPs feel their voice is not heard		
Theme three	HCPs struggle to redirect women's decision regarding mode of birth	HCPs choose not to redirect women's decision HCPs have difficulty redirecting when there is a window to do so Sometimes it is too late to redirect		

bit of medical paternalism and perhaps led her into a situation... that later we all realised that a patient who does not want to have a vaginal birth is not going to have a vaginal birth (Hospital 5, obstetrics resident).

On the other hand, changes from outside the health services include the approval and regulation of the Law of Humanised Birth, which is believed to allow choice of mode of birth in clinical settings (and therefore request a CS without medical indication), and changes in the attitudes of pregnant woman and their companions of choice. 'Patients have changed,' state HCPs, and they underpin this transformation in women's empowerment, their being more informed regarding their rights and the influence of media.

I think that women are now quite informed in general, because with all the discussion about the media, the laws, in other words, they are more involved (Hospital 5, obstetrician).

# Theme 2: HCPs feel powerless against women's request to choose mode of birth

Even though HCPs claim to have adopted—or be trying to adopt—a new paradigm regarding decision-making, they sometimes feel frustrated when women want to make decisions on mode of birth. Patient-centered decision-making seems to be an acceptable option provided that women choose the option that HCPs, especially doctors, consider appropriate. HCPs state that 'the decision belongs to the woman'; however, pregnant women and families are many times portrayed as making uninformed 'wrong' decisions regarding mode of birth. Fear of litigation in case of adverse neonatal outcomes is presented as a reason to conduct a CS against HCPs' own judgement, especially when it concerns 'pushy' women or companions. In these cases, CS appears to be the only option for some HCPs:

There are patients who want a caesarean section at all costs, and no matter how much you... you work under a lot of pressure, patients have changed a lot, there is a lot of pressure, sometimes you have patients who pressure you and say "do the caesarean section, if something happens to me it will be your responsibility...," and even if you don't want to, at the slightest doubt you are going to have a caesarean section (Hospital 5, midwife).

In the case of CS without medical indication, HCPs state that they have to accept women's or companions' decision regarding mode of birth even if it is not the best clinical option for the patient due to the Law of Humanized Birth. Their 'voice' and expertise are no longer taken into account in the matter, as women are protected by law to make "unilateral" decisions. As a result, HCPs feel powerless:

Because if it [the cesarean section] is by law, it is put "by law" and the pregnant woman's right is protected by that law (...). By law, we have no say in anything, not as professionals or anything else. The patient has already made the decision and you have to operate, and the problem is over, your speciality is over, just like that (Hospital 2, obstetrician).

HCPs consider that the wording of the regulation of Law 25,929 of Humanised Birth now allows women to choose mode of birth. This has had consequences in clinical practice, and HCPs report to be performing CSs upon maternal request that are not medically indicated:

Law 25,929 in its regulation, in the last paragraph of the regulation, those who interpreted it, misinterpreted it... it was different varieties of positions, or that the woman could give birth in any position, and they put mode of birth [vía de determinación], so the woman can choose caesarean section. Today I have already had (...) three caesarean sections by Law 25,929, two nulliparous, which were totally unnecessary, because we didn't do any trial of labour, or anything, but legally they are allowed (Hospital 2, head of obstetric service).

HCPs state that some pregnant women also push for a VB when it is not a safe clinical option. In these cases, HCPs do not accept women's choice, since it might result in an adverse maternal outcome for which they would be legally responsible:

We have patients with birth plans, in which the patient at all costs wanted a vaginal birth, a vaginal birth, a vaginal birth. However, she had two caesarean sections before, so no, I told her, not here. If you want that, go somewhere else, not here, and she had to accept, because two caesarean sections, the truth is that I am not going to do a vaginal birth (Hospital 3, head of obstetric service).

The complexities of the decision-making process, as well as the underlying power struggle, makes itself evident when what is questioned it is not the mode of birth that a woman chooses but rather *who* makes the decision and on what basis. HCPs claim that many women make decisions regarding mode of birth without being informed about the risk and benefits of the procedures, even in cases when women state a concrete reason for wanting a CS, for example, not wanting to cope with labour pain.

Especially in the hospital, you see that we deal with a lower middle class, they are uninformed, and I think that in recent times it is more... they refuse the information that you can give them and they already come with a concept that they do not want pain and they choose a caesarean section, that is, they come with that already, that information in their heads (Hospital 4, obstetrician).

The issues presented above suggest that HCPs consider *themselves* to be the ones who should be in control of the process of decision-making in the process of labour and birth, both in case of a CS or a VB. This assessment collides with their own accounts of women's rights and agency.

But they come with the little piece of paper [the birth plan], they don't let you touch her, they don't let you (...) you have to accept it, what do you want me to say? Like the mothers who now don't want to vaccinate their children, I don't agree, but... in public health they shouldn't be allowed to do that (Hospital 3, head of obstetric service).

It is worth noticing, however, that some midwives in our sample expressed a different point of view than most HCPs (both obstetricians and other midwives) regarding women choosing mode of birth, suggesting, for example, that women choose a CS as part of the empowerment the Law of Humanised Birth aims to enhance:

Yes, it can be a conditioning factor for a mother who is empowered by the Law on Respectful Childbirth to say, "I decide that I want to have a caesarean section" (Hospital 1, midwife).

# Theme 3: HCPs struggle to redirect women's decision regarding mode of birth

HCPs mention three paths in their clinical practice when dealing with CS upon maternal request without medical indication: They sometimes choose not to redirect women's decision; they sometimes try to change her mind by presenting the risks and benefits of each type of birth, but they struggle to do so; sometimes they consider that it is too late to redirect for legal reasons.

In the first case, HCPs state that the fact that women use the Law of Humanised Birth to ask for a CS, although its original purpose was to protect women against unnecessary interventions. The law is sometimes used as an "excuse" by HCPs to perform CSs, which are viewed as a kind of "easy way out":

The patient is also informed about the law, which I think is good, but at the same time (...) the purpose of the law was different, but as it is not clear, they

use it the other way round, the purpose of the law was to avoid unnecessary interventions, but as it says that the woman can choose, she can also choose a caesarean section, so in this aspect they [HCPs] take advantage of the situation and say "well, if the woman doesn't want it [a vaginal birth], she doesn't want it," and it ends up being a caesarean section (Hospital 5, midwife).

Paradoxically, HCPs also emphasize the importance of explaining to women the risk of caesarean section and the process of labour, so that they may change their mind about asking for an unnecessary CS. HCPs mention that communication with women is something they struggle with, and that it is not always possible to get them to change their mind. When HCPs frame this dilemma focusing on changing women's mind instead of helping them to make an informed decision, it illustrates a wish to prescribe that is part of the paternalistic mindset they claim to have left behind.

We spend a lot of time talking to patients, we spend more and more time, we criticise those who don't do it, we even make patients sign a consent form stating that they have been informed of the indication for caesarean section, in order to see if we can lower the rate a little, but what kills us, kills us, is the refusal to try labour (Hospital 4, obstetrician).

According to HCPs, the reasons why women who demand a CS will not change their minds are manifold. HCPs suggest that women being more informed about their right to choose mode of birth does not mean that they are more aware of the risks and benefits of CS and VB. Not all women access or attend prenatal courses, which results in lack of preparation during pregnancy. Miscommunication between providers and pregnant women on what to expect during the process of labour and misinformation during antenatal care were also mentioned as important factors. Some healthcare professionals also mention that have lost their prestige in the eyes of patients, and therefore their opinions are sometimes not valued.

It is very difficult to get [pregnant women] to come to the childbirth preparation classes, so sometimes they arrive without knowing what the birth is going to be like, they are often not prepared for the birth, and they also sometimes have a bad opinion of doctors and the [medical] institution. It is something that happens in general, the doctor has lost prestige, and our specialty has lost the admiration of the patients (Hospital 4, obstetrician). Finally, sometimes it is too late to redirect women's request for a CS for legal reasons. Healthcare providers report that due to the Law of Humanised birth, women can ask for a CS when being admitted in the hospital. This decision is written in the medical record of the women, and HCPs cannot attempt to change it:

That law protects pregnant woman's right and obviously we have no other option, because the decision has already been made and put in the clinical record, so we don't.... We are not involved in any kind of decision; the decision has already been made by the patient (Hospital 2, obstetric resident).

#### Discussion

Choice of mode of birth is a complex process influenced by a vast array of issues. Our analysis raised the importance of power struggles between HCPs in decision making on the mode of birth. While providers claim to believe that they have changed previous paternalistic practices, allowing pregnant women to become an active actor in decision-making, they feel frustrated when they are not able to influence women's choice of mode of birth. We have conceptualized this as loss of *beneficial power* [29]. Paradoxically, providers sometimes use women's request for a CS as an excuse to perform a surgical birth, since it appears to be "easier" for them. This suggests that the interviewees are not against performing CS per se, but rather that they want to retain control over decision-making.

The loss of *beneficial power* is also experienced by HCPs when women or their companions question their knowledge and prestige. In the traditional biomedical model of care, doctors were the ones to define the terms and boundaries of the discussions with patients, given their legitimised expertise [17, 28, 30]. This provided them with more control over the decision-making process. Interviewed HCPs, especially medical doctors, perceive their authority as being challenged and have trouble making women and family "listen" to their medical advice. A previous qualitative study has similarly suggested that HCPs find shared-decision making to be hindered when patients request medical practices that do not align with HCPs experiences or knowledge [26].

In the last few decades, CS at women's request has been a key aspect of the controversy regarding decisionmaking on mode of birth. A recent mixed-methods study conducted in Argentina on the views of obstetricians, midwives and residents in public healthcare facilities showed that there was no agreement among professional groups on whether CS upon request is a determinant of the CS increase in this country [48]. Although there is no updated information about the proportion of CS due to maternal request, our study suggests that it appears to play an important role in the doctor-patient interaction regarding mode of birth. The Law of Humanised Birth has a central place in HCPs accounts regarding decisionmaking on mode of birth. HCPs believe that, because of the way it is written, the law has in practice allowed for choice on mode of birth. This, in turn, appears to have had a considerable impact on power relations among HCPs, pregnant women and companions and the way decisions are made in the clinical settings. More research is needed to understand how the Law of Humanised Birth is currently interpreted in maternity wards regarding choice of mode of birth.

HCPs feel powerless upon women being legally able to ask for a CS without medical indication. They mention fear of litigation as a significant reason for agreeing to perform a CS upon request, especially when pregnant women or family members put pressure on them. This has also been pointed out by previous literature as an important determinant of CS increase [53]. A recent study in Mexico also showed that HCPs felt powerless and frustrated upon women and families asking for a CS, especially when faced with difficult or aggressive patients [24].

Women's lack of preparation during pregnancy and misinformation during antenatal care regarding what to expect during the labour process were suggested as important reasons why pregnant women refuse to try a VB, especially based on not wanting to cope with labour pain. This is in line with previous literature, which has shown that fear of pain during labour is a reason for women to prefer a CS [9, 54–58]. Recent quantitative studies on women's preferences on mode of birth in Argentina and Brazil indicated that avoiding pain was one of the main reasons put forward by women preferring a CS to a VB [47, 59]. An important contextual factor to take into consideration is that only one out of the five participating hospitals provides epidural analgesia during labour as a routine practice.

HCPs also mention misinformation about the risks and benefits of each mode of birth as a contributing factor for CS upon maternal request. The aforementioned study from Argentina showed that the main reason for women preferring a CS was that they perceived it as a safer option than VB [47], which is in line with research conducted in other countries [57, 60–63] but contradicts the available evidence on the risks and benefits of each mode of birth [14, 64]. According to the interviewees, women are also influenced by media, which has been identified as having a significant role in agenda setting for fashionable trends regarding mode of birth [65]. Finally, women look for meaningful communication and interaction with their HCPs when deciding on mode of birth [66, 67]. Our study showed that miscommunication between pregnant women and HCPs is a factor that plays a role in the power dynamics regarding mode of birth decision-making. HCPs emphasize the importance of explaining the risks and benefits of each type of birth to women, yet they report that they do not always find it easy to interact with them. More research is needed to understand why HCPs in Argentina find it difficult to establish a fruitful conversation with pregnant and labouring women in healthcare facilities.

The literature on the topic suggests that a shared process of decision-making [29] may be the best way to disentangle the complex issue of choosing mode of birth [9, 29, 55, 65, 68]. Evidence has suggested that women being in control and able to choose during the process enhances the birth experience [69-71]. However, certain conditions must be met for the decision to be taken together by HCPs and pregnant women. Taking into account women's opinions, values and desires, as well as respecting their autonomy and agency has been suggested as paramount [29]. Reliable and updated information provided to women in healthcare facilities that will allow them to weigh the risks and benefits of each mode of birth has been considered essential [57, 66, 72]. Moreover, meaningful communication and interaction with HCPs has been shown to be an important factor for women when making a decision on mode of birth, and therefore should be consider a fundamental aspect of shared decisionmaking [19, 66, 73]. Finally, provision of emotional support before and during labour has also been shown to help women in the decision-making process [66, 67].

#### Limitations

This qualitative study used the audio recording and qualitative interview transcripts of a formative research conducted in Argentina [44], collected with a different aim and using a guide that was not exclusively tailored to our research questions. Moreover, data saturation was decided in line with the purpose of the formative research. Even though data was not collected with the purpose of looking at power relations, this issue was addressed in several parts of the interviews. For example, during this data-driven analysis, we were able to compare how power was expressed at different stages: antenatal care, admission, labour, and birth. Moreover, it was a topic tackled by different types of HCPs (OBGYN specialists, midwives and residents) in all participating hospitals. It should be mentioned that only three of the five included hospitals had midwives in their staff, hindering the comparison between different cadres per hospital.

We report on very specific power relations between HCPs and pregnant women in five public maternity wards in Argentina in the context of a new legal scenario regarding mode of birth. Transferability to other contexts includes public hospitals in this country where statistics on CS are available, as well as other contexts with similar legislation or in which the practice of CS upon medical request has been a concerning issue. We also consider the results of this study to potentially be of use to researchers, decision-makers, policy makers and HCPs in settings that are trying to shift from a doctor-centered towards a patient-centered model of maternity care.

#### Conclusion

Decision-making on mode of birth remains an intricate debate that should not only look into medical research findings, women's and providers' autonomy, ethics, law and media, but also into clinical practice and the interaction between HCPs, women and companions. This study suggests that power relations within the healthcare setting and the legal framework in which these relations take place are important dimensions to take into account in the mode of birth controversy. Our analysis highlights the complexity of the HCP-patient interaction in the context of a Law of Humanised Birth that, in practice, is understood as allowing women to choose mode of birth. Even though HCPs claim to be changing their biomedical mindset, welcoming women to be an active part of the decision-making processes in labour and birth, they feel frustrated and powerless when women make autonomous decisions regarding mode of birth, especially in the case of CS without medical indication. HCPs perceive themselves to be losing *beneficial power* and prestige in the eyes of patients, and that fruitful communication on risks and benefits of each mode of birth is not always possible in the mode of birth decision-making process. Furthermore, HCPs perform more and more CS without medical indication when it is convenient for them, which suggests that paternalistic practices and beliefs are still in place.

#### Abbreviations

CS Caesarean section HCP Healthcare provider OBGYN Obstetrician/gynaecologist VB Vaginal birth

#### Acknowledgements

We thank Silvina Ramos, Mariana Romero, Carla Perrota, Yanina Sguassero, Cecilia Straw and Natalia Righetti, who developed and carried out this formative research. We also thank Alexandre Dumont and the QUALI-DEC team for creating a conducive research environment on appropriate use of C-sections.

#### Author contributions

All authors contributed in this article. MVO coded the interviews, developed the theoretical framework, built the themes and led the writing of the manuscript. HMA oversaw each step of the process. MVO held several rounds of discussion with HMA and CG to develop and refine the themes. CG also provided insight about the clinical practice in the context of the study. CH, APB and GC discussed and agreed on the final themes and provided meaningful insights regarding the mode of birth controversy and relevant aspects of clinical practice in maternity wards. All co-authors reviewed the manuscript and provided significant intellectual content. All authors agreed on the final version of the manuscript.

#### Funding

Open access funding provided by Karolinska Institute. The UNDP/UNFPA/ UNICEF/WHO/World Bank Special Programme of Research, Development and Research Training in Human Reproduction (HRP), a cosponsored program executed by the World Health Organization (WHO) in the Department of Sexual and Reproductive Health and Research (SRH), funded this project, in addition to the Swedish Research Council (project number 2020–04949). The contents of this article are solely the responsibility of the authors.

#### Availability of data and materials

Data are available on reasonable request.

#### Declarations

#### Ethics approval and consent to participate

The project "Optimizing the use of caesarean section in Argentina: Design and methodology of a formative research for the development of interventions" was approved by the Independent Ethics Committee of Centro Rosarino de Estudios Perinatales (record number 2/18, January 2018). It was also approved by the provincial Ethics Committees and/or the Teaching and Research Committees at each of the selected hospitals according to the requirements of each jurisdiction. The Research Project Review Panel of the UNDP/UNFPA/ UNICEF/WHO/World Bank Special Programme of Research, Development and Research Training in Human Reproduction (HRP) at the Department of Sexual and Reproductive Health and Research of WHO and the WHO Research Ethics Review Committee, Geneva, Switzerland have also approved this project. In Argentina, the research protocol was registered in the RENIS database (number IS002316). The participants consented to participate prior to the interview. The person in charge of conducting the interviews obtained informed consent. Participants were given the opportunity to read the form aloud and ask questions prior to signing the consent. The informed consent form included information on the study and stated that the participation in the study was voluntary and that participants were free to skip any question and withdraw from the study at any point without consequences.

#### **Consent for publication**

Not applicable.

#### **Competing interests**

The authors declare no competing interests.

#### Author details

<sup>1</sup>Department of Global Public Health, Karolinska Institutet, Stockholm, Sweden. <sup>2</sup>Facultad de Ciencias de la Salud Blanquerna, Universidad Ramón Llull, Barcelona, Spain. <sup>3</sup>Centro Rosarino de Estudios Perinatales, Rosario, Argentina. <sup>4</sup>UNDP/UNFPA/UNICEF/WHO/World Bank Special Programme of Research, Development and Research Training in Human Reproduction, Department of Sexual and Reproductive Health and Research, World Health Organization, Geneva, Switzerland.

#### Received: 13 May 2023 Accepted: 7 August 2023 Published online: 21 August 2023

#### References

- 1. Esen UI, Archibong EI. Maternal request caesarean section: maternal vs medical autonomy. Br J Hosp Med. 2007;68(11):610–1.
- Klein VR, McDowl J. Should pregnant women be able to choose elective cesarean as a birth option? MCN Am J Matern Child Nurs. 2010;35(5):252–3.

- Chervenak FA, McCullough LB, Brent RL. The professional responsibility model of obstetrical ethics: avoiding the perils of clashing rights. Am J Obstet Gynecol. 2011;205(4):315.e1-315.e5.
- Kapfhamer JD, Menon S, Spellecy R. The importance of risk tolerance in maternal autonomy. Am J Bioeth. 2012;12(7):53–4.
- Gee H. Caesarean section should be available on request: FOR: the mother's autonomy should be paramount. BJOG. 2015;122(3):359.
- Silver RM. Caesarean section should be available on request: AGAINST: caesarean delivery on maternal request is a bad idea. BJOG Int J Obstet-Gynaecol. 2015;122(3):360.
- Loke AY, Davies L, Mak Y-w. Is it the decision of women to choose a cesarean section as the mode of birth? A review of literature on the views of stakeholders. BMC Pregnancy Childbirth. 2019;19(1):286.
- ACOG Committee Opinion No. 761: cesarean delivery on maternal request. Obstet Gynecol. 2019;133(1):e73–7.
- Sorrentino F, Greco F, Palieri T, Vasciaveo L, Stabile G, Carlucci S, et al. Caesarean section on maternal request-ethical and juridic issues: a narrative review. Medicina. 2022;58(9):1255.
- 10. Jomeen J. The paradox of choice in maternity care. J Neonatal Nurs. 2012;18(2):60–2.
- 11. Chervenak F, McCullough LB. FOR: responding professionally to requests for non-indicated caesarean delivery. BJOG. 2015;122(3):360.
- Betrán AP, Ye J, Moller AB, Zhang J, Gülmezoglu AM, Torloni MR. The increasing trend in caesarean section rates: global, regional and national estimates: 1990–2014. PLoS ONE. 2016;11(2):e0148343.
- Betrán AP, Ye J, Moller AB, Souza JP, Zhang J. Trends and projections of caesarean section rates: global and regional estimates. BMJ Glob Health. 2021;6(6):e005671.
- Souza JP, Gülmezoglu AM, Lumbiganon P, Laopaiboon M, Carroli G, Fawole B, et al. Caesarean section without medical indications is associated with an increased risk of adverse short-term maternal outcomes: the 2004–2008 WHO global survey on maternal and perinatal health. BMC Med. 2010;8(1):71.
- Berg J, Källén K, Andolf E, Hellström-Westas L, Ekéus C, Alvan J, et al. Economic evaluation of elective cesarean section on maternal request compared with planned vaginal birth-application to swedish setting using national registry data. Value Health. 2023;26(5):639–48.
- Betran AP, Torloni MR, Zhang JJ, Gülmezoglu AM, Section WHOWGoC. WHO statement on caesarean section rates. BJOG. 2016;123(5):667–70.
- 17. Lupton D. Medicine as culture: illness, disease and the body. London; 2012. https://sk.saqepub.com/books/medicine-as-culture-3e.
- Kaimal AJ, Kuppermann M. Understanding risk, patient and provider preferences, and obstetrical decision making: approach to delivery after cesarean. Semin Perinatol. 2010;34(5):331–6.
- Doraiswamy S, Billah SM, Karim F, Siraj MS, Buckingham A, Kingdon C. Physician-patient communication in decision-making about caesarean sections in eight district hospitals in Bangladesh: a mixed-method study. Reprod Health. 2021;18(1):34.
- Kaimal AJ, Kuppermann M. Decision making for primary cesarean delivery: the role of patient and provider preferences. Semin Perinatol. 2012;36(5):384–9.
- Chadwick R. Bodies that birth: vitalizing birth politics. London: Routledge; 2018.
- Beckett K. Choosing cesarean: feminism and the politics of childbirth in the United States. Fem Theory. 2005;6(3):251–75.
- 23. Kitzinger S. Sheila Kitzinger's letter from Europe: the politics of birth. Birth. 2003;30(3):203–5.
- Smith-Oka V, Flores B. Competing narratives: examining obstetricians' decision-making regarding indications for cesarean sections and abdominal incisions. Soc Sci Med. 2022;309:115238.
- Keedle H, Schmied V, Burns E, Dahlen HG. A narrative analysis of women's experiences of planning a vaginal birth after caesarean (VBAC) in Australia using critical feminist theory. BMC Pregnancy Childbirth. 2019;19(1):142.
- Diamond-Brown L. It can be challenging, it can be scary, it can be gratifying": obstetricians' narratives of negotiating patient choice, clinical experience, and standards of care in decision-making. Soc Sci Med. 2018;205:48–54.
- Lagerløv P. Claiming power in doctor-patient talk; by Nancy Ainsworth-Vaughn. Oxford University Press, Oxford, 1998. 212 pp. Soc Sci Med. 2000;50:1514.

- Kaba R, Sooriakumaran P. The evolution of the doctor–patient relationship. Int J Surg. 2007;5(1):57–65.
- Eide KT, Bærøe K. How to reach trustworthy decisions for caesarean sections on maternal request: a call for beneficial power. J Med Ethics. 2020;47(12):e45.
- Foucault M. In: Power/knowledge: selected interviews and other writings, 1972–1977. Brighton: Harvester Press; 1980.
- Fleming P, Spicer A. Power in management and organization science. Acad Manag Ann. 2014;8(1):237–98.
- Benoit C, Zadoroznyj M, Hallgrimsdottir H, Treloar A, Taylor K. Medical dominance and neoliberalisation in maternal care provision: the evidence from Canada and Australia. Soc Sci Med. 2010;71(3):475–81.
- World Health Organization. Regional Office for the Western P. Peoplecentred health care: a policy framework. https://apps.who.int/iris/ handle/10665/206971.
- Drife J. The start of life: a history of obstetrics. Postgrad Med J. 2002;78(919):311–5.
- 35. Foucault M. The history of sexuality: first American edition. New York: Pantheon Books; 1978.
- Brubaker SJ, Dillaway HE, Medicalization. Natural childbirth and birthing experiences. Sociol Compass. 2009;3(1):31–48.
- Neiterman E. Sharing bodies: the impact of the biomedical model of pregnancy on women's embodied experiences of the transition to motherhood. Healthc Policy. 2013;9(Spec Issue):112–25.
- Wade DT, Halligan PW. Do biomedical models of illness make for good healthcare systems? BMJ. 2004;329(7479):1398–401.
- Silva G, Moura M, Queiroz A, Pereira A, Carvalho A, Netto L. Possibilidades para a mudança do modelo obstétrico hegemônico pelas enfermeiras obstétricas. Revista Enfermagem UERJ. 2020;28:e49421.
- Shaw D, Guise JM, Shah N, Gemzell-Danielsson K, Joseph KS, Levy B, et al. Drivers of maternity care in high-income countries: can health systems support woman-centred care? Lancet. 2016;388(10057):2282–95.
- Ley 25929 parto humanizado y decreto 2035/15. Ministerio de Justicia y Derechos Humanos de la Nación. Secretaría de Derechos Humanos y Pluralismo Cultural. 2018. https://www.argentina.gob.ar/sites/default/ files/ley\_25929\_parto\_humanizado\_decreto\_web\_0.pdf.
- Kaplan Westbrook L. El parto humanizado: perspectivas de profesionales en las maternidades públicas de Buenos Aires. Independent Study Project (ISP) Collection. 2015;2245.
- Binfa L, Pantoja L, Ortiz J, Cavada G, Schindler P, Burgos RY, et al. Midwifery practice and maternity services: a multisite descriptive study in Latin America and the Caribbean. Midwifery. 2016;40:218–25.
- 44. Ramos S, Romero M, Perrotta C, Sguassero Y, Straw C, Gialdini C, et al. Optimizing the use of cesarean section in Argentina: design and methodology of a formative research for the development of interventions. Reprod Health. 2021;18(1):23.
- Harrison MS, Garces A, Figueroa L, Esamai F, Bucher S, Bose C, et al. Caesarean birth by maternal request: a poorly understood phenomenon in low- and middle-income countries. Int Health. 2021;13(1):63–9.
- 46. Dumont A, Betrán AP, Kaboré C, de Loenzien M, Lumbiganon P, Bohren MA, et al. Implementation and evaluation of nonclinical interventions for appropriate use of cesarean section in low- and middle-income countries: protocol for a multisite hybrid effectiveness-implementation type III trial. Implement Sci. 2020;15(1):72.
- 47. Perrotta C, Romero M, Sguassero Y, Ingram C, Righetti N, Gialdini C, et al. Women's mode of birth preferences and preparedness of hospitals to support vaginal birth in the public health sector in Argentina. Reprod Female Child Health. 2022;1(2):111–21.
- Perrotta C, Romero M, Sguassero Y, Straw C, Gialdini C, Righetti N, et al. Caesarean birth in public maternities in Argentina: a formative research study on the views of obstetricians, midwives and trainees. BMJ Open. 2022;12(1):e053419.
- Ministerio de Salud de la República Argentina. Dirección de Estadísticas e Información de Salud. https://www.argentina.gob.ar/salud/deis.
- 50. Adams W. Conducting semi-structured interviews. Oxford: Oxfam; 2015.
- Braun V, Clarke V. Using thematic analysis in psychology. Qual Res Psychol. 2006;3:77–101.
- Braun V, Clarke V. Reflecting on reflexive thematic analysis. Qual Res Sport Exerc Health. 2019;11(4):589–97.

- Sarah E, Elena A, Soo D, Joanna E, Sunny M, Gill M, et al. Behavioural factors associated with fear of litigation as a driver for the increased use of caesarean sections: a scoping review. BMJ Open. 2023;13(4):e070454.
- Jenabi E, Khazaei S, Bashirian S, Aghababaei S, Matinnia N. Reasons for elective cesarean section on maternal request: a systematic review. J Matern Fetal Neonatal Med. 2020;33(22):3867–72.
- Coates D, Thirukumar P, Spear V, Brown G, Henry A. What are women's mode of birth preferences and why? A systematic scoping review. Women Birth. 2020;33(4):323–33.
- Loke AY, Davies L, Li S-F. Factors influencing the decision that women make on their mode of delivery: the Health Belief Model. BMC Health Serv Res. 2015;15(1):274.
- 57. Betrán AP, Temmerman M, Kingdon C, Mohiddin A, Opiyo N, Torloni MR, et al. Interventions to reduce unnecessary caesarean sections in healthy women and babies. Lancet. 2018;392(10155):1358–68.
- Fuglenes D, Aas E, Botten G, Øian P, Kristiansen IS. Why do some pregnant women prefer cesarean? The influence of parity, delivery experiences, and fear. Am J Obstet Gynecol. 2011;205(1):45.e1-45.e9.
- 59. Domingues RM, Dias MA, Nakamura-Pereira M, Torres JA, d'Orsi E, Pereira AP, et al. Process of decision-making regarding the mode of birth in Brazil: from the initial preference of women to the final mode of birth. Cad Saude Publica. 2014;30(Suppl 1):1–16.
- Torloni MR, Betrán AP, Montilla P, Scolaro E, Seuc A, Mazzoni A, et al. Do italian women prefer cesarean section? Results from a survey on mode of delivery preferences. BMC Pregnancy Childbirth. 2013;13(1):78.
- Pang MW, Lee TS, Leung AK, Leung TY, Lau TK, Leung TN. A longitudinal observational study of preference for elective caesarean section among nulliparous Hong Kong Chinese women. BJOG. 2007;114(5):623–9.
- Hug I, Chattopadhyay C, Mitra GR, Kar Mahapatra RM, Schneider MC. Maternal expectations and birth-related experiences: a survey of pregnant women of mixed parity from Calcutta, India. Int J Obstet Anesth. 2008;17(2):112–7.
- Dweik D, Girasek E, Töreki A, Mészáros G, Pál A. Women's antenatal preferences for delivery route in a setting with high cesarean section rates and a medically dominated maternity system. Acta Obstet Gynecol Scand. 2014;93(4):408–15.
- 64. Ye J, Zhang J, Mikolajczyk R, Torloni MR, Gülmezoglu AM, Betran AP. Association between rates of caesarean section and maternal and neonatal mortality in the 21st century: a worldwide population-based ecological study with longitudinal data. BJOG. 2016;123(5):745–53.
- Duckworth S. Should maternal choice be an indication for caesarean section? Int J Surg. 2008;6:277–80.
- Kingdon C, Downe S, Betran AP. Non-clinical interventions to reduce unnecessary caesarean section targeted at organisations, facilities and systems: systematic review of qualitative studies. PLoS ONE. 2018;13(9):e0203274.
- 67. Ojelade OA, Titiloye MA, Bohren MA, Olutayo AO, Olalere AA, Akintan A, et al. The communication and emotional support needs to improve women's experience of childbirth care in health facilities in Southwest Nigeria: a qualitative study. Int J Gynecol Obstet. 2017;139(S1):27–37.
- Shareef N, Scholten N, Nieuwenhuijze M, Stramrood C, de Vries M, van Dillen J. The role of birth plans for shared decision-making around birth choices of pregnant women in maternity care: a scoping review. Women Birth. 2022;36:327–33.
- Cook K, Loomis C. The impact of choice and control on women's childbirth experiences. J Perinat Educ. 2012;21(3):158–68.
- Hallam JL, Howard CD, Locke A, Thomas M. Communicating choice: an exploration of mothers' experiences of birth. J Reprod Infant Psychol. 2016;34(2):175–84.
- Green JM, Baston HA. Feeling in control during labor: concepts, correlates, and consequences. Birth. 2003;30(4):235–47.
- Klein MC. Cesarean section on maternal request: a societal and professional failure and symptom of a much larger problem. Birth. 2012;39(4):305–10.
- Nieuwenhuijze MJ, Korstjens I, de Jonge A, de Vries R, Lagro-Janssen A. On speaking terms: a Delphi study on shared decision-making in maternity care. BMC Pregnancy Childbirth. 2014;14(1):223.

Content courtesy of Springer Nature, terms of use apply. Rights reserved.

#### **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

**M. Vila Ortiz** Department of Global Public Health, Karolinska Institutet, Stockholm, Sweden; Centro Rosarino de Estudios Perinatales, Rosario, Argentina. She is a social scientist working as a researcher in sexual and reproductive health at Centro Rosarino de Estudios Perinatales and a Ph.D. student at the Department of Global Public Health at Karolinska Institutet.

**C. Gialdini** Facultad de Ciencias de la Salud Blanquerna-Universidad Ramón Llull, Barcelona, Spain; Centro Rosarino de Estudios Perinatales, Rosario, Argentina. She is an OBGYN working clinically in Argentina and as a researcher at Centro Rosarino de Estudios Perinatales, so she is familiar with the context of the study.

**C. Hanson** Department of Global Public Health, Karolinska Institutet, Stockholm, Sweden. She is a medical doctor with a specialisation in gynaecology and obstetrics, with a wide experience working internationally on issues of maternal and perinatal health.

**A. P. Betrán** UNDP/UNFPA/UNICEF/WHO/World Bank Special Programme of Research, Development and Research Training in Human Reproduction, Department of Sexual and Reproductive Health and Research, World Health Organization, Geneva, Switzerland. She is a medical doctor with expertise in research on public health issues of global relevance to maternal and perinatal health.

**G. Carroli** Centro Rosarino de Estudios Perinatales, Rosario, Argentina. He is an obstetrician and researcher with a wide experience working on randomized controlled trials and large global surveys on the main causes of maternal and perinatal morbidity and mortality.

**H. Mølsted Alvesson** Department of Global Public Health, Karolinska Institutet, Stockholm, Sweden. She is a medical anthropologist with expertise in qualitative research in health, working with global health challenges.

#### Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

#### At BMC, research is always in progress.

Learn more biomedcentral.com/submissions



### Terms and Conditions

Springer Nature journal content, brought to you courtesy of Springer Nature Customer Service Center GmbH ("Springer Nature").

Springer Nature supports a reasonable amount of sharing of research papers by authors, subscribers and authorised users ("Users"), for smallscale personal, non-commercial use provided that all copyright, trade and service marks and other proprietary notices are maintained. By accessing, sharing, receiving or otherwise using the Springer Nature journal content you agree to these terms of use ("Terms"). For these purposes, Springer Nature considers academic use (by researchers and students) to be non-commercial.

These Terms are supplementary and will apply in addition to any applicable website terms and conditions, a relevant site licence or a personal subscription. These Terms will prevail over any conflict or ambiguity with regards to the relevant terms, a site licence or a personal subscription (to the extent of the conflict or ambiguity only). For Creative Commons-licensed articles, the terms of the Creative Commons license used will apply.

We collect and use personal data to provide access to the Springer Nature journal content. We may also use these personal data internally within ResearchGate and Springer Nature and as agreed share it, in an anonymised way, for purposes of tracking, analysis and reporting. We will not otherwise disclose your personal data outside the ResearchGate or the Springer Nature group of companies unless we have your permission as detailed in the Privacy Policy.

While Users may use the Springer Nature journal content for small scale, personal non-commercial use, it is important to note that Users may not:

- 1. use such content for the purpose of providing other users with access on a regular or large scale basis or as a means to circumvent access control;
- 2. use such content where to do so would be considered a criminal or statutory offence in any jurisdiction, or gives rise to civil liability, or is otherwise unlawful;
- 3. falsely or misleadingly imply or suggest endorsement, approval, sponsorship, or association unless explicitly agreed to by Springer Nature in writing;
- 4. use bots or other automated methods to access the content or redirect messages
- 5. override any security feature or exclusionary protocol; or
- 6. share the content in order to create substitute for Springer Nature products or services or a systematic database of Springer Nature journal content.

In line with the restriction against commercial use, Springer Nature does not permit the creation of a product or service that creates revenue, royalties, rent or income from our content or its inclusion as part of a paid for service or for other commercial gain. Springer Nature journal content cannot be used for inter-library loans and librarians may not upload Springer Nature journal content on a large scale into their, or any other, institutional repository.

These terms of use are reviewed regularly and may be amended at any time. Springer Nature is not obligated to publish any information or content on this website and may remove it or features or functionality at our sole discretion, at any time with or without notice. Springer Nature may revoke this licence to you at any time and remove access to any copies of the Springer Nature journal content which have been saved.

To the fullest extent permitted by law, Springer Nature makes no warranties, representations or guarantees to Users, either express or implied with respect to the Springer nature journal content and all parties disclaim and waive any implied warranties or warranties imposed by law, including merchantability or fitness for any particular purpose.

Please note that these rights do not automatically extend to content, data or other material published by Springer Nature that may be licensed from third parties.

If you would like to use or distribute our Springer Nature journal content to a wider audience or on a regular basis or in any other manner not expressly permitted by these Terms, please contact Springer Nature at

onlineservice@springernature.com